Author's response to reviews

Title: What is preventable harm in healthcare? A Systematic Review of definitions

Authors:

Mohammed Nabhan (mhmdnab85@gmail.com)
Tarig Beraima Elraiyah (elraiyah.tarig@mayo.edu)
Daniel R Brown (brown.daniel@mayo.edu)
James A Dilling (jdilling@mayo.edu)
Annie LeBlanc (leblanc.annie@mayo.edu)
Victor M Montori (montori.victor@mayo.edu)
Timothy I Morgenthaler (tmorgenthaler@mayo.edu)
James M Naessens (naessens@mayo.edu)
Larry J Prokop (prokop.larry@mayo.edu)
Veronique L Roger (roger.veronique@mayo.edu)
Stephen J Swensen (swensen.stephen@mayo.edu)
Rodney L Thompson (thompson.rodney@mayo.edu)
M. Hassan Murad (murad.mohammad@mayo.edu)

Version: 2 Date: 5 December 2011

Author's response to reviews:

Dear editor,

We would like to submit our research paper entitled, “What is preventable harm in healthcare? A Systematic Review of definitions” for publication to BMC HSR. Our systematic review aims at surveying the medical literature for existing and emerging definitions of preventable harm.

Thank you for your consideration.